Author’s response to reviews

Title: Transcatheter arterial chemoembolization plus sorafenib versus transcatheter arterial chemoembolization alone to treat advanced hepatocellular carcinoma: A meta-analysis

Authors:

Rong Cai (570039478@qq.com)
Rongfeng Song (929766813@qq.com)
Yan Yan (ronger2003011327@163.com)
Yifeng Liao (815354896@qq.com)
Cuiling Zhou (1536559823@qq.com)
Shuncong Wang (1624481793@qq.com)
Xiuling Zhou (987709137@qq.com)
Huaping Wang (344189537@qq.com)
Hongyu Zhang (zdwyzhy@163.com)
Haiqing Ma (mahaiqing@mail.sysu.edu.cn)
Huanhuan Sun (sunhuanh3@mail.sysu.edu.cn)
Pengfei Pang (785535164@qq.com)

Version: 2 Date: 01 Aug 2017

Author’s response to reviews:

Editor Comments: Please make a major revision according to the reviewers' comments. Reviewer reports: If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage.

Response: Thanks for your suggestion and advices. To polish our revised manuscript, we have asked American Journal Experts (http://bit.ly/AJE_BS) to help us in English editing.